Author’s response to reviews

Title: A randomised controlled trial to test the effectiveness of decision training on assessors' ability to determine optimal fitness-to-drive recommendations for older or disabled drivers.

Authors:

Priscilla Harries (priscilla.harries@brunel.ac.uk)

Carolyn Unsworth (c.unsworth@cqu.edu.au)

Hulya Gokalp (Hulya.Gokalp@brunel.ac.uk)

Miranda Davies (miranda.davies@nuffieldtrust.org.uk)

Christopher Tomlinson (chris.tomlinson@imperial.ac.uk)

Luke Harries (lh511@cam.ac.uk)

Version: 2 Date: 21 Jan 2018

Author’s response to reviews:

Dear Ms Mollmuth

Thank you for confirming the final four points below prior to acceptance:

1. In the Ethics and consent to participate section please clarify regarding consent to participate. If consent was not required, please include a statement on whether the need for consent was waived by an IRB or if national guidelines have established that studies such as this don't require consent. Response: Informed consent was obtained online by firstly asking participants to read the participant information text and then to consent via completing a tick box before access to participation online could commenced.

2. Please include a statement in Authors’ contributions clearly indicating that all authors read and approved the final version of the manuscript. Response: This has been added.

3. Please remove the information about trial registration from your abstract since the trial registration was not necessary. It is sufficient to provide this information in the manuscript. Response: This has been removed.

4. Please ensure that when you upload your revised submission that it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Response: This has been done.